Author’s response to reviews

Title: Small Scale Migration along the Interoceanic Highway in Madre de Dios, Peru: An exploration of community perceptions and dynamics due to migration

Authors:

Kelly Jensen (jensenke90@gmail.com)
Nehal Naik (nnaik27@gmail.com)
Christina O'Neal (coneal3@tulane.edu)
Gabriela Salmón-Mulanovich (gabriela.salmon.mulanovich@gmail.com)
Amy Riley Powell (amyrileypowell@gmail.com)
Gwennyth Lee (gwennyth.lee@gmail.com)
Stella Hartinger Peña (stella.hartinger.p@upch.pe)
Daniel Bausch (Daniel.Bausch@phe.gov.uk)
Valerie Paz Soldan (vpazsold@tulane.edu)

Version: 2 Date: 30 Jan 2018

Author’s response to reviews:

Dear the editorial committee,

The requested changes have been made the manuscript. The authorship form includes handwritten signatures. Figure titles and legends have been included in the manuscript and consent methods have been clarified (written consent was obtained from all participants). A copy of the interview guide has been included as well.

Best,

Kelly Jensen